Author's response to reviews

Title: How are "teaching the teachers" courses in evidence based medicine evaluated? A systematic review

Authors:

Jacek Walczak (jacek.walczak@caspin.org.pl)
Anna Kaleta (a.kaleta@caspin.org.pl)
Elżbieta Gabry (egabrys@caspin.org.pl)
Krzysztof Kloc (kkloc@caspin.org.pl)
Shakila Thangaratinam (s.thangaratinam@bham.ac.uk)
Gemma Barnfield (g.barnfield@bham.ac.uk)
Susanne Weinbrenner (weinbrenner@azq.de)
Berit Meyerrose (meyerrose@azq.de)
Theodores N Arvanitis (t.arvanitis@bham.ac.uk)
Andrea R Horvath (ahorvath@clab.szote.u-szeged.hu)
Gianni Zanrei (gianni.zanrei@unicatt.it)
Regina Kunz (rkunz@uhbs.ch)
Katja Suter (suterka@uhbs.ch)
Bernard Burnand (Bernard.Burnand@chuv.ch)
Chantal Arditi (Chantal.Arditi@chuv.ch)
Katrien Oude Rengerink (K.OudeRengerink@amc.uva.nl)
Harry Gee (harry.gee@bwhct.nhs.uk)
Ben WJ Mol (b.w.mol@amc.uva.nl)
Khalid S Khan (k.s.khan@bham.ac.uk)

Version: 2 Date: 10 August 2010

Author's response to reviews:

Dear Editor

Many thanks for accepting our manuscript for publication. We have carefully looked into the reviewer's comments and made revisions. I have submitted the revised manuscript.

Regards

Shakila Thangaratinam